\documentclass{exam}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{graphicx,wasysym,gensymb}
\usepackage[hmargin=1in,tmargin=.75in,bmargin=1in]{geometry}
\usepackage[dvipsnames,table]{xcolor}
\usepackage{leftidx}
\usepackage{pgfplots}
\pagestyle{empty}

\usepackage{amsthm}
\makeatletter
\newtheoremstyle{indented}
  {.15cm}% space before
  {.15cm}% space after
  {\addtolength{\@totalleftmargin}{3.5em}
   \addtolength{\linewidth}{-3.5em}
   \parshape 1 3.5em \linewidth}% body font
  {}% indent
  {\bfseries}% header font
  {.}% punctuation
  {.5em}% after theorem header
  {}% header specification (empty for default)
\makeatother
\theoremstyle{indented}
\newtheorem{exmp}{Example}
\newtheorem{prac}{Practice}

\usepackage{tikz}
\usepackage{tikz-qtree}
\usetikzlibrary{calc,shapes,arrows.meta}

\usepackage{grffile}

\usepackage{enumitem}

\usepackage{stackengine}

\usepackage{pdfpages}

\usepackage{blkarray, bigstrut}

\usepackage{caption}

\usepackage{pifont}

\newcommand\funderline[1]{\stackengine{0pt}{\underline{\qquad\qquad\vphantom{#1}}}{#1}{O}{c}{F}{F}{L}}

\newcommand\tline[2]{\underset{\text{#1}}{\funderline{$#2$}}}

\newcommand\myperm[2]{\leftidx{_#1}{P}{_#2}}

\newcommand\mycomb[2]{\leftidx{_#1}{C}{_#2}}

\newcommand*\NewPage{\newpage\null}

\pgfmathdeclarefunction{gauss}{3}{\pgfmathparse{1/(#3*sqrt(2*pi))*exp(-((#1-#2)^2)/(2*#3^2))}}

\newcommand\CircN[1]{%
  \tikz[baseline=(char.base)]\node[circle,draw,inner sep=2pt] (char) {#1};}

\newcommand\CircP[1]{%
  \tikz[baseline=(char.base)]\node[circle,draw,inner sep=4.75pt] (char) {#1};}
  
\newcommand\BLUE[1]{\textcolor{blue}{\textbf{#1}}}
\newcommand\WHITE[1]{\textcolor{white}{\textbf{#1}}}
\newcommand{\PreserveBackslash}[1]{\let\temp=\\#1\let\\=\temp}
%\newcolumntype{C}[1]{>{\PreserveBackslash\centering}p{#1}}
%\newcolumntype{R}[1]{>{\PreserveBackslash\raggedleft}p{#1}}
%\newcolumntype{L}[1]{>{\PreserveBackslash\raggedright}p{#1}}
\usepackage{array}
\newcolumntype{C}[1]{%
 >{\vbox to 1in\bgroup\vspace{0.1in}\centering}%
 p{#1}%
 <{\egroup}}  
\newcommand{\cmark}{\ding{51}}
\newcommand{\xmark}{\ding{55}}

\usepackage[%
  colorlinks=true,%
  urlcolor=blue%
]{hyperref}



\begin{document}
\newpage
%\setcounter{page}{1}
\noindent Name: \rule{7cm}{.2pt} \vspace{.3cm} \hfill Date:  \rule{3.5cm}{.2pt}
\vspace{.3cm}

\noindent\hrulefill
\begin{center}
{\bf \Large{MA 162}} \\\vspace{.1cm} Week 1 Recitation Worksheet
\end{center}
\vspace{-.3cm}
\hrulefill
\vspace{.3cm}


\raggedright
\noindent\textbf{Directions:}  Below is a table consisting of 25 traits/descriptions.  Walk around and introduce yourself to your fellow classmates and TA.  As you do so, try to find people who match the traits/descriptions in the table.  When you find someone who matches a trait/description, write their name (first name plus first letter of last name) in the corresponding box.  You must fill in at least 10 of the boxes, without listing anyone more than once, to earn full credit for this week.

\begin{center}
\begin{tabular}{| C{1in} | C{1in} | C{1in} | C{1in} | C{1in} |}
\hline
Was born outside of Kentucky & Is adventurous & Has met someone famous & Agrees that Chris Hemsworth is the superior `Chris' & Has a tattoo \tabularnewline \hline
Enjoys playing video games & Has ridden a horse & Is left-handed & Is the youngest or only child & Enjoys reading books \tabularnewline \hline
Binge-watched Tiger King during the pandemic & Played a sport in high school & Went to a concert last year & Loves Olive Garden & Has traveled outside the US \tabularnewline \hline
Saw a movie during the break & Is not the first in their family to attend UK & Has milked a cow & Has ran at least a 5K & Prefers Coke over Pepsi \tabularnewline \hline
Plays a musical instrument & Enjoys cooking or baking & Enjoys watching (Keeping Up With) The Kardashians & Has a pet & Is fluent in at least 2 languages \tabularnewline \hline
\end{tabular}
\end{center}
\end{document}